Author’s response to reviews

Title: Continuous intravenous infusion of remifentanil improves the experience of parturient undergoing repeated cesarean section under epidural anesthesia, a prospective, randomized study

Authors:

wei yan (m18768389432@163.com)
Yun Xiong (jxeyxy@126.com)
Yu Yao (weiqiuair@hotmail.com)
Feng-jiang Zhang (fengjiang_zhang@126.com)
Li-an Yu (zryulina@zju.edu.cn)
Min Yan (zryanmin@zju.edu.cn)

Version: 4 Date: 30 Nov 2019

Author’s response to reviews:

Dear editor,

First of all, we thank both reviewers and editor for their positive and constructive comments and suggestions.

According to the comments of editor, we have revised the manuscript, and responded, point by point to, the comments as listed below.

I would like to re-submit this revised manuscript “Continuous intravenous infusion of remifentanil improves the experience of parturient undergoing repeated cesarean section under epidural anesthesia, a prospective, randomized study” (BANE-D-19-00032R3) to BMC Anesthesiology, and hope it is acceptable for publication in the journal.

Looking forward to hearing from you soon.

With kindest regards,

Yours sincerely,

Min Yan. M.D.
Professor and Chief of Anesthesiology Department
The Second Affiliated Hospital of Zhejiang University School of Medicine
88 Jiefang Road, Hangzhou, China
Editor Comments:
1. Trial registration
   -- Unfortunately I am still unable to return any registered trial with the identifier 'ChiCRT1800018423' (see attached image). Without the correct trial registration number we are unable to accept your paper for publication, please to provide the correct identifier. The BMC-series journals does accept retrospectively registered trials. If you have not registered the trial, we therefore request that you do so as soon as possible so that your study can be accepted for publication.
   Response : Thank you for your hard work. We are sorry for the mistake. The correct register number is “ChiCTR1800018423” and we have revised it in our manuscript. See page 2.
2. Raw data
   -- As additional file 1 contains raw data, and therefore does not contain critical information for understanding your study, please remove it from your manuscript to ensure that participant anonymity is not compromised.
   At your discretion, you may instead state the raw data available upon reasonable request from the corresponding authors in the “Availability of data and materials” section of the declarations.
   Response : Thanks. In current submission, we have remove the raw data. And we have stated that the datasets used and/or analysed during the current study available from the corresponding author on reasonable request in the revised manuscript. See page 12.